Author’s response to reviews

Title: Serum phosphate and social deprivation independently predict all-cause mortality in chronic kidney disease.

Authors:

Marit Solbu (maritdaso@gmail.com) (marit.solbu@unn.no) (Marit.Dahl.Solbu@glasgow.ac.uk)

Peter Thomson (peter.thomson@nhs.net)

Sarah Macpherson (sarah.macpherson2@nhs.net)

Mark Findlay (Mark.Findlay@glasgow.ac.uk)

Kathryn Stevens (Kate.Stevens@glasgow.ac.uk)

Rajan Patel (Rajan.Patel@glasgow.ac.uk)

Sandosh Padmanabhan (Sandosh.Padmanabhan@glasgow.ac.uk)

Alan Jardine (Alan.Jardine@glasgow.ac.uk)

Patrick Mark (Patrick.Mark@glasgow.ac.uk)

Version: 2 Date: 22 Oct 2015

Author’s response to reviews:

Dear Editor,

Thank you very much for your e-mail this morning to inform us that our article “Serum phosphate and social deprivation independently predict all-cause mortality in chronic kidney disease” has been accepted for publication in your journal. We are certainly pleased to receive this good news.

In this revised version, we have specified each author's contribution to the article simply by adding all the authors' initials in the final sentence of the “Authors' Contributions” paragraph.

No further changes have been made.

We hope this minor change has been done to your satisfaction. Please do not hesitate to contact the corresponding author if you need further changes to be made.